# THEORETICAL ADVANCES

# A novel feature-based tracking approach to the detection, localization, and 3-D reconstruction of internal defects in hardwood logs using computer tomography

Suchendra M. Bhandarkar · Xingzhi Luo · Richard Daniels · E. William Tollner

Received: 18 March 2005/Accepted: 11 February 2006/Published online: 18 August 2006 © Springer-Verlag London Limited 2006

Abstract A novel feature-based tracking approach based on the Kalman filter is proposed for the detection, localization, and 3-D reconstruction of internal defects in hardwood logs from cross-sectional computer tomography (CT) images. The defects are simultaneously detected, classified, localized, and reconstructed in 3-D space, making the proposed scheme computationally much more efficient than existing methods where the defects are detected and localized independently in individual CT image slices and the 3-D reconstruction of the defects accomplished via correspondence analysis across the various CT image slices. Robust techniques for defect detection and classification are proposed. Defect class-specific tracking schemes based on the Kalman filter, B-spline contour approximation, and Snakes contour fitting are designed which use the geometric parameters of the defect contours as the tracking variables. Experimental results on cross-sectional CT images of hardwood logs from select species such as white ash, hard maple, and red oak are presented.

S. M. Bhandarkar (⊠) · X. Luo Department of Computer Science, The University of Georgia, Athens, GA 30602-7404, USA e-mail: suchi@cs.uga.edu

R. Daniels Warnell School of Forest Resources, The University of Georgia, Athens, GA 30602-2152, USA

E. W. Tollner

Department of Biological and Agricultural Engineering, The University of Georgia, Athens, GA 30602-4435, USA **Keywords** Computer tomography · Internal defect detection · Kalman filter · Feature-based tracking · Nondestructive evaluation · Automated lumber production

# 1 Original contribution of the work

The paper describes the design and implementation of a novel feature-based tracking approach for the detection, localization, and 3-D reconstruction of internal defects in hardwood logs from cross-sectional computer tomography (CT) images. With declining trends in hardwood forest resources and increasing costs of hardwood logs, improving the lumber value vield via detection and localization of internal log defects prior to lumber production has become imperative for many sawmills. Although the technical feasibility of CT scanning of logs in real time is fast approaching reality, the computational methods for analyzing the resulting CT images reliably and in real time are still a major challenge. In most conventional approaches, the internal log defects are detected and localized by analyzing individual CT image slices independently and the 3-D reconstruction of the defects accomplished via correspondence analysis across the various CT image slices. The resulting computational complexity makes the conventional approaches unsuitable for real-time applications. In the proposed scheme, defect detection, defect localization, and 3-D defect reconstruction are integrated within a Kalman filter-based feature tracking framework by exploiting spatial coherence across successive CT image slices. Robust techniques for defect detection and classification are proposed. Defect class-specific tracking schemes based on the Kalman filter, B-spline contour approximation, and Snakes contour fitting are designed which use the geometric parameters of the defect contours as the tracking variables. Robust techniques for extraction and characterization of the external log surface are also designed. Since the internal defects and the external log surface are simultaneously detected, classified, localized, and reconstructed in 3-D space, the proposed scheme is shown to be a major improvement over conventional approaches in terms of both computational efficiency and detection accuracy. Experimental validation of the proposed scheme on cross-sectional CT images of hardwood logs from select species such as white ash, hard maple, and red oak is also presented.

# 2 Introduction

The value of hardwood lumber is determined by the quantity, size, and types of internal log defects such as knots, cracks, decay, and other anomalies of tree growth that eventually appear on the lumber surfaces. Depending on the nature of the hardwood end utilization, each log must be sawed to minimize the presence of these internal defects on the resulting lumber surfaces. In order to achieve this goal, the internal defects within the log must be accurately identified and localized prior to the sawing of the log. In most sawmills, however, logs are processed into lumber with little or no information about the internal log defects and with inaccurate or incomplete geometric data about the logs. The logs are sawed by a sawmill operator who uses only his best judgement based on external log inspection and knowledge of lumber grades to make sawing decisions. Sometimes the sawmill operator unknowingly chooses the best sawing pattern for a log. However, many times his choices are far from optimal and less lumber or lumber of a lower grade is produced. Furthermore, boredom and fatigue of the sawmill operator can affect the accuracy of the lumber processing, resulting in suboptimal lumber production where the potential value of logs is wasted.

Production of lumber is essentially a destructive process. As the log is sawed, new information is divulged on the quality of the wood inside which may suggest a different and better sawing pattern. However, since each step in the sawing process is irreversible, the loss in the value yield has already been incurred and cannot be rectified. Haygreen [1] has reported a low conversion efficiency of about 35% for conventional sawmills. With the rising costs of hardwood logs accounting for over 80% of total production costs [2], improving the lumber value yield from hardwood logs has become imperative for many sawmills. Given the inherent limitations of external log inspection, it is reasonable to assume that future gains in lumber value yield will be possible only by internal log scanning [2–10].

Identification and localization of internal log defects are estimated to lead to potential gains of about 15-18% in lumber value, representing a savings of over \$2 billion for the hardwood lumber industry in the United States [10, 11]. Forest products-based economies are increasingly dependent on getting the highest value wood products from a declining forest resource base. This results in disproportionate harvesting pressure on high-demand hardwood species such as hard maple, black walnut, white ash, and red oak that exhibit the greatest differences in value between the highest and lowest lumber grades [12]. Environmental concerns and the ecological need for maintaining biodiversity in forest ecosystems underscore the need to utilize as many hardwood species for wood products as possible, to improve the efficiency in converting low-grade logs into high-value lumber products, to reduce unnecessary wastage, and to conserve valuable hardwood forest resources. One way of achieving these goals is by identification and localization of internal defects in hardwood logs and using this information to optimize the processing of the resulting lumber.

Studies of computer axial tomography (CAT or CT) and magnetic resonance imaging (MRI) [also known as nuclear magnetic resonance (NMR) imaging] for internal log defects [2, 6-8, 10, 13-15] have demonstrated that the CT and MRI technologies available today can be used successfully to image the internal features of logs. CT scanners which are essentially solid state (i.e., with a minimum of moving parts) can scan at rates exceeding 30 slices per second. Thus, the technical feasibility of scanning logs in real time is fast approaching reality. However, the computational methods for analyzing the CT images for internal defects reliably and in real time, and exploiting the knowledge of the internal defects to determine optimal lumber processing strategies remain a challenging and open research topic.

In this paper we describe the design and implementation of a computer vision system for the detection, localization, and 3-D reconstruction of internal defects in hardwood logs from cross-sectional CT images. Robust techniques for defect detection and classification are proposed. In contrast to traditional methods, where the defects are detected and localized independently in individual CT image slices and the 3-D reconstruction of the defects accomplished via correspondence analysis across the various CT image slices, the proposed system integrates defect detection, defect localization, and 3-D defect reconstruction within a Kalman filter-based feature tracking framework. The defects are simultaneously detected, classified, localized, and reconstructed in 3-D space, making the proposed system computationally much more efficient than existing systems. Defect classspecific tracking schemes based on the Kalman filter are designed and implemented. The Kalman filter [16] uses the geometric parameters of the defect contours identified in the individual CT image slices as the tracking variables. The pixels corresponding to the outer boundary of the log cross-section are extracted in each CT image slice and collated across all the CT image slices to generate the external surface of the log. The identified and localized internal defects and the external surface of the log are reconstructed in 3-D space, resulting in a virtual 3-D model of the log along with its internal defects. The 3-D model of the log and internal defects could subsequently be used by a computer-based lumber production planning system to determine an optimal sawing strategy. Alternatively, the system could be used as a decision aid by sawyers and machinists in a sawmill where the detection and visualization of the internal defects and computer simulation of key machining operations would assist in determining an optimal lumber production strategy for a given hardwood log. The system could also be used as an interactive training tool for novice sawyers and machinists whereby they could practice various sawing strategies on virtual logs before working on real logs.

The remainder of the paper is organized as follows. Section 3 provides a brief review of previous work. Section 4 provides an overview of the proposed computer vision system. Section 5 provides a detailed scheme for the detection, identification, and localization of internal defects in a single CT image slice using a combination of structural and spectral features. The procedures for the detection, identification, and localization of knots, holes, and cracks in a single CT image slice are detailed in Sects. 5.1, 5.3, and 5.4, respectively. Section 6 describes the 3-D reconstruction of the internal defects using Kalman filter-based tracking algorithms. The 3-D reconstruction procedures for knots, and for cracks and holes are presented in Sects. 6.1 and 6.2, respectively. Section 7 presents experimental results on real CT image data from hardwood logs. Section 8 concludes the paper with an outline for future work.

# 3 A brief review of previous work

Computer axial tomography (CT or CAT) and MRI [also known as NMR imaging] represent two potentially viable techniques for acquiring cross-sectional images of logs. Although MRI is a more recent innovation, CT technology is approaching the speed necessary for real-time production use in sawmills [2]. MR images are characterized by the fact that wet portions of the object being imaged appear as relatively light regions in the resulting image, whereas dry portions of the object appear as relatively dark regions in the image. Hence MRI techniques are particularly well suited for detecting internal features of logs, such as knots, reaction wood, wet wood, and gum spots that are characterized by varying moisture content in the underlying wood [3, 17].

In CT images, the grayscale value of a pixel is directly proportional to the X-ray absorption which is then correlated with the material density at the pixel location [2, 8, 15, 18]. Knots and moisture pockets are noted to have higher material density and/or higher moisture content than surrounding clear wood and are often characterized by pixels with very high grayscale values in the CT image. Holes and decay pockets are void areas that are filled with air or decayed wood, respectively, and hence characterized by low material density. Consequently, the CT image pixels corresponding to holes and decay pockets have very low grayscale values. Cracks are also filled with air, and like holes and decay pockets, are characterized by pixels with very low grayscale values in the CT image. The major difference is that holes and decay pockets usually have circular cross-sections and are typically short in length whereas cracks are usually thin and long.

Gray-level thresholding and binarization techniques have been used extensively in internal defect identification in cross-sectional CT images of logs. As depicted in (1), the grayscale value of a pixel F(i,j) is used to classify the pixel as belonging to one of the major internal defect classes such as knots, moisture pockets, holes, decay pockets and cracks [4–7, 10, 19–22].

 $Defect_Class(i, j)$ 

$$= \begin{cases} \text{Knots, moisture pockets} & \text{if } F(i,j) > T_1 \\ \text{Cracks, holes, decay pockets} & \text{if } F(i,j) < T_2 \\ \text{Normal wood} & \text{otherwise} \end{cases}$$
(1)

In (1), the threshold values  $T_1$  and  $T_2$  are determined by analyzing the grayscale histogram over all the CT image slices. The grayscale thresholding operation

)

is followed by region growing that groups spatially contiguous pixels with the same class label into defect regions. As can be seen from (1), a grayscale-based classification scheme cannot distinguish between knots and moisture pockets or between cracks, holes, and decay pockets. Neural networks have been used to integrate grayscale features within a local 2-D and 3-D pixel neighborhood for internal defect classification [23–25]. However, since only gray level values are used as inputs, the neural network is reported to have problems distinguishing between knots and moisture pockets, both of which are characterized by high grayscale values in the CT images [24, 25]. Consequently, more recent approaches use shape and texture features in addition to pixel grayscale values to identify and localize the major internal defects [13, 14, 26–29]. Zhu et al. [29] use the basic 3-D geometric features of the defects in conjunction with the Dempster-Schafer theory of evidential reasoning for defect classification. Butler et al. [22, 26] impose the spatial coherence constraint in their defect classification algorithms. Samson [30] presents a geometrical model to describe knots within logs and on the surface of lumber beams sawn from those logs. Samson [30] also presents an algorithm to compute the effect of the presence of knots in the conversion of logs into structural lumber. Bhandarkar et al. [13, 14] use the 3-D geometry of the reconstructed defects for the purpose of classification. In particular, the 3-D shape parameters are used to distinguish between knots and moisture pockets and between cracks and holes/decay pockets.

The results of internal defect identification and localization can be used to reconstruct a 3-D model of the log along with its internal defects [13, 14, 21, 31–34]. Software programs that simulate various machining operations such as sawing and veneering on the virtual 3-D log reconstruction have been described in the literature [13, 14, 35–38]. Likewise, programs for automated grading of virtually produced lumber (using the aforementioned simulators) and physical lumber produced by sawmills have also been reported in the literature [4, 39–44]. These programs have been used to estimate the improvements in value yield recovery for lumber production using internal log scanning which are reported to be in the range of 40–60% [14, 36, 45].

The above review covers the most relevant developments in the use of CT and MRI technologies for internal defect detection and identification in hardwood logs. The review paper by Pham and Alcock [46] covers the wider area of automated visual inspection of logs and lumber and includes both external scanning using optical sensors and internal scanning using CT and MRI technologies.

#### 4 Overview of the proposed defect detection system

The proposed computer vision system for identification, localization, and 3-D reconstruction of internal defects in hardwood logs via analysis of cross-sectional CT images is motivated by two major limitations of most existing systems. First, most existing computer vision systems identify internal log defects using thresholding algorithms that are based on the analysis of the pixel-level grayscale histogram computed from the input CT or MR image. However, it has been observed that such pixel-level thresholding or binarization methods are often inadequate in terms of their ability to discriminate the defect-containing areas in the input image from the defect-free areas even with fairly sophisticated analysis of the grayscale histogram [13, 14, 35–38]. In many CT images, the grayscale values of the pixels comprising a knot are usually close to those of pixels comprising the bright rings in the annular ring structure of the log. Consequently, if the value of threshold  $T_1$  in (1) is chosen to be too high, some pixels comprising a true knot might be mistakenly classified as non-knots. Conversely, with a very low value for threshold  $T_1$ , many pixels comprising the bright rings will be misclassified as comprising a knot. Figure 1a shows an example CT image containing two knots (i.e., the two homogeneous bright areas). With a threshold value  $T_1 = 185$ , the region classified as a knot in the upper right corner of the image contains too many pixels which actually belong to the ring structure (Fig. 1b). Simultaneously, some pixels which actually comprise the knot are classified as non-knots. A similar problem exists in the case of detection of cracks. A very low value for threshold  $T_2$  in (1) results in fragmentation of the crack, whereas too large a value for  $T_2$  results in the inclusion of pixels comprising the grayscale valleys between successive bright rings in the log ring structure. Figure 2a shows a CT image containing a crack. With a threshold value  $T_2 = 50$ , the binarized image (Fig. 2b) can be seen to include several pixels comprising the grayscale valleys between successive rings in the log ring structure. Fragmentation of the crack can also be observed in Fig. 2b. These problems are further exacerbated when the log is too wet or too dry. When the log is too wet, there is a greater overlap between the grayscale values of pixels comprising the ring structure and those of pixels comprising the knots. Conversely, when the log is too dry, the grayscale values of pixels comprising the holes and cracks are almost the same as those of pixels comprising the grayscale valleys between successive rings in the log ring structure. The above examples clearly demonstrate the need to combine structural or

and the result of knot



Fig. 2 A example CT image and the result of crack detection using (1): a original CT image, and **b** binary image

shape information along with spectral or grayscale information within the defect detection and identification process.

Second, in most existing computer vision systems, the detection, identification, and localization of internal log defects are done independently in each CT or MR image slice. Simple correspondence analysis across multiple CT image slice is performed to reconstruct the defect in 3-D space [13, 14]. The geometrical parameters pertaining to the defect shape, size, and location are used to establish the correspondence across multiple image slices. However, this technique is computationally inefficient and unsuitable for real-time application since the spatial coherence of the defect along the axial direction is not exploited. In addition, since only two successive slices are used to establish the correspondence, the reconstruction is prone to error in the case of defects with complex 3-D shapes. Clearly, a technique that more closely integrates the processes of defect detection, identification, and 3-D reconstruction is needed.

The proposed system addresses both the aforementioned shortcomings. Defect detection and identification is achieved by integrating the spectral and structural properties of potential defect-containing regions in a CT image slice. Knots and rings share the same grayscale values but differ in their structural properties. Rings can be modeled as narrow strip-like regions in a local window, whereas knots can be modeled as ellipsoidal regions with bounded eccentricity. Analysis of the pixel grayscale density in a local window centered at each pixel within the defect-containing region is used to distinguish between the knots and rings. Since the above technique tends to cause erosion of the knot boundaries, a morphological operator is used to recover the boundaries of the knots. Since cracks are typically thin and long, they are also modeled as narrow strip-like regions in a local window. The grayscale valleys between successive rings also exhibit the same grayscale values as the cracks. However, the fact that a crack is usually perpendicular to the rings and grayscale valleys is exploited [47]. The

points of intersection of a crack with the rings and grayscale valleys, termed as fork points, are detected and grouped to extract the line segments comprising the crack. False crack lines are removed via analysis of orientation of these line segments.

The processes of detection, identification, and reconstruction of the log and its internal defects are integrated within a Kalman filtering framework. A Kalman filter [16] is used to track cracks and holes continuously from one CT image slice to the next. A dynamic contour extraction technique based on Kalman Snakes [48] is used to reconstruct the knots and the outer boundary of the log in 3-D space. The Kalman filter is used to predict the locations of the defects in successive slices. This results in much faster detection, identification, and localization of the defects since only a local search within a fairly small neighborhood of the predicted defect locations is entailed. For example, when the approximate location and shape of a knot contour in the current CT image slice are known, the Kalman Snakes method can be used to predict the location and shape of the new contour in a succeeding CT image slice using local information, thus speeding up the overall computation. The overall flowchart for the proposed defect detection system is given in Fig. 3.

# 5 Defect detection in a single CT image

The two primary goals of the proposed computer vision system are (1) to integrate structural and spectral features in the processes of detection, localization, and 3-D reconstruction of internal defects and (2) to integrate the processes of detection, localization, and 3-D reconstruction of internal defects within a single Kalman filter-based feature tracking framework. Defect



Fig. 3 The flowchart of the proposed defect detection system

detection and localization within a single CT image slice is performed in order to initialize the Kalman filter-based tracking algorithm. After a defect is detected and localized within a CT image slice, the tracking algorithm is used to detect and localize the defect in successive CT image slices and also reconstruct it in 3-D space.

# 5.1 Detection of knots

Since a simple pixel-level binarization or thresholding scheme typically misclassifies the bright annular ring areas as knots, a new binarization method is proposed to separate the knot-containing regions from the ring areas. Knots and rings share similar grayscale values but differ in their structural properties. Rings can be modeled as narrow strip-like regions in a local window, whereas knots can be modeled as ellipsoidal regions with bounded eccentricity. The ring areas can be eliminated via analysis of the pixel grayscale density in a local window centered at each pixel within the defect-containing region as outlined in (2).

$$K(i,j) = \begin{cases} 1 & \text{if } \frac{1}{M^2} \sum_{(x,y) \in W(i,j)} F(x,y) > T_k \\ 0 & \text{otherwise} \end{cases}$$
(2)

where K(i,j) is the resulting binary image delineating the knots, W(i,j) is a window of predetermined size  $M \times M$  centered at pixel (*i*,*j*), F(i,j) is a CT image slice of the log, and  $T_k$  is a predetermined threshold used to classify a pixel as a knot pixel. The binary image resulting from the application of (2) to the CT image slice in Fig. 1a is shown in Fig. 4a. A comparison of Figs. 1a and 4a shows that the application of (2) results in the erosion of the knot boundaries. Consequently, a morphological dilation operation [49] is used to recover the knot boundaries. The morphological dilation operation is represented as  $K \oplus B$  where  $\oplus$  is the dilation operator, K is the binary image resulting from the application of (2), and B is the dilation operator mask of size  $M \times M$  depicted in Fig. 4b. The knots extracted after applying the dilation operation are shown in Fig. 4c.

#### 5.2 Detection of the outer log boundary

The outer boundary of the log cross-section in a single CT image slice can be simply detected using the binarization technique given in (2). However, the size of the window W used for the detection of the outer log boundary is much smaller (3 × 3 is typically adequate). Moreover, the threshold value T used for detection of



the outer log boundary is also much smaller; typically it is slightly larger than the grayscale value of a void area in the CT image.

(a)

# 5.3 Detection of holes

Holes are actual void areas and appear as dark regions in the CT images with graylevels similar to those of the background. A simple thresholding scheme outlined in (3) is used to classify the CT image pixels as holes.

$$K(i,j) = \begin{cases} 1 & \text{if } F(i,j) < T_{h} \\ 0 & \text{otherwise} \end{cases}$$
(3)

Since holes are usually small in size and approximately round in shape, false holes, typically caused by small cracks or grayscale valleys between successive rings, are removed by using a combination of morphological erosion and dilation operations on the thresholded result [49]. The mask size for the erosion and dilation operators in the case of detection of holes is smaller than the mask size of the dilation operator used in knot detection. A  $3 \times 3$  mask size is used for the erosion and dilation operators for hole detection. Figure 5 depicts the procedure to detect holes. Figure 5a is a CT image slice which contains a crack and several holes. After thresholding, holes are retained in the resulting binary image. However, a number of nonhole regions are also detected (Fig. 5b). After application of the erosion operator (Fig. 5c), the nonhole pixels are removed, but the holes are eroded. The dilation operator is then applied to restore the holes to their true size (Fig. 5d).

5.4 Crack detection

A crack, as it appears in a CT image of a log, is usually long and thin. A straightforward grayscale-based binarization of the image results in either fragmentation of the detected cracks or too many misclassifications of regions denoting the grayscale valleys between the annular rings as cracks (as shown in Fig. 2b). Since both the grayscale valleys and cracks are narrow and long, the grayscale density-based binarization technique (2) is not able to separate them. However, cracks are typically perpendicular to the grayscale valleys and the local direction of a grayscale valley can be estimated by approximating the grayscale valleys by concentric circles centered at the centroid of the log crosssection. This property is exploited in the proposed crack detection scheme.

The proposed crack detection scheme is summarized as follows. The local linear structures resulting from cracks and valleys are first detected. Since the lines defining the cracks and the lines defining the local structure of the valleys intersect each other, the points of intersection are represented as *fork* points within a local window. By detecting and localizing these fork points, the actual crack features can be detected and localized. In order to reject false crack features, which are usually caused by the presence of random noise between the rings, it is determined whether or not the detected crack feature is parallel (within a certain angular threshold) to the local structure of the valleys. If the detected crack line is indeed parallel to the local structure of the grayscale valleys, then it is discarded else it is retained as an actual crack feature.

Fig. 5 Result of hole detection: **a** input CT image, **b** thresholded image, **c** removal of cracks and valleys using erosion, and **d** restoration of holes using dilation



In order to detect the edges associated with the crack features, four directional Sobel-like operators are used. These operators perform a combination of direction-sensitive smoothing and edge detection for linear features. These operators are observed to reliably detect and localize the linear features in a CT image, even in the presence of noise. They can detect crack features without causing them to be fragmented. Additionally, the grayscale valleys between the rings are detected as well. A sample image resulting from the application of these operators is shown in Fig. 6a. The resulting image is binarized using a simple thresholding technique and a thinning (i.e., skeletonization) algorithm described in [50] is used to skeletonize the binary image (Fig. 6b). Fork points in the thinned image are detected and localized using a local  $7 \times 7$  window. Fork points that are distributed along the same crack feature are grouped together using a greedy clustering algorithm that is based on connectivity and mutual proximity of the fork points. The clustering algorithm exploits the fact that fork points on the same crack feature should be spatially proximate and connected to each other. Groups of fork points that are parallel to the local ring or valley structure are discarded. Figure 6c, d shows the results of fork detection and fork grouping, respectively.

The groups of fork pixels that are retained are characterized by line segments that describe the crack feature. The RANSAC algorithm [51] is used to perform line fitting since it is robust to the presence of outliers in the input data. The RANSAC algorithm randomly samples a subset of points from the given set of data points to perform least-squares line fitting. The data points whose distances from the fitted line are above a certain threshold are considered as outliers, whereas the remaining data points are considered to be inliers. If the number of inliers is above a certain percentage of the total number of data points, the least-squares line fitting is performed again but the input data points are restricted to the inlier set. If the inlier set is not large enough, the above procedure is repeated on a new subset of the input data points.

In our case, the RANSAC-based line fitting procedure is not restricted to the fork pixels since the number of candidate fork pixels in a group is typically not large enough to ensure robust line fitting. Hence, in addition to the fork pixels within a group, the pixels between the fork pixels in the skeletonized (i.e., thinned) image are considered as well. In order to achieve this, the candidate fork pixels within a group are ordered. The bounding box for the group of candidate fork pixels is computed. If the width of the bounding box is larger than the height, the candidate fork pixels in the group are ordered along the x axis, otherwise they are ordered along the y axis. If  $\{p_i\}$  is the ordered set of candidate fork pixels within the group, then for each pair of successive fork pixels  $(p_i, p_{i+1})$ , a local window can be defined by using the pixels  $(p_i, p_{i+1})$  as the corner points of the window. The 1-pixels in the skeletonized image obtained from within all the windows defined for every pair of successive candidate fork pixels together with the fork pixels are used in the line fitting procedure. Figure 7 depicts the result of the RANSAC-based line fitting procedure.

The extraction of the actual crack line segment in the binary image is based on the parameters of the fitted line. The primary reason for doing so is that the line segment representing the actual crack line may be longer than the line segment defined by the fork pixels. This is so because the ring structure is weak in the vicinity of the center of the log cross-section, causing fork pixels to go undetected. Thus, a 1-pixel in the binary image is deemed to belong to the actual crack line segment, if it is both, close to the fitted line and also spatially connected via a path consisting entirely of 1-pixels to the fork pixels within the group. An iterative depth-first search procedure detailed in [52] is used to localize the actual crack pixels.

# 6 Three-dimensional defect reconstruction using the Kalman filter

Internal defects within the log are observed to exhibit spatial coherence across several successive CT image slices, i.e., the appearances of an internal defect in

Fig. 6 Result of crack detection: **a** output of the Sobel-like operators, **b** result of thinning the binary image, **c** result of fork detection, and **d** result of fork grouping





Fig. 7 The result of the RANSAC-based line fitting procedure

successive CT image slices share similar attributes, such as location, shape, size, and grayscale distribution. Once an internal defect is detected in an image slice, its attributes in successive slices can be predicted. Thus, when the attributes of an internal defect in previous image slices are known, it is much more efficient to predict its attribute values in succeeding image slices and then compute the actual attribute values by using the predicted values as the starting point. Defect correspondence between successive slices is automatically established in this scheme thus precluding the need to perform correspondence analysis independently for each pair of successive image slices. The result is a computationally more efficient defect extraction and 3-D defect reconstruction procedure. The Kalman filter is adopted as the prediction scheme for this purpose. Computationally efficient algorithms for detection of knots, holes, and cracks are designed to take advantage of the predicted defect attribute values resulting from the Kalman filter. Continuous defect tracking and 3-D construction across CT image slices are achieved by iteratively repeating the process of Kalman filter-based prediction and fast defect extraction.

Knots, holes, and cracks exhibit very different geometrical attributes in a 2-D image slice and hence have different Kalman filtering models. A knot can be approximated by a homogeneous region bounded by a convex contour. The convexity assumption enables efficient geometric computation. A crack is usually long and thin and is approximated by a line segment. A hole usually has a smaller cross-sectional area and can be represented by a rectangular bounding box. In addition, the outer boundary of the log cross-section is also detected and tracked across CT image slices in order to virtually reconstruct the entire log in 3-D space. The knots and the outer boundary of the log crosssection can be simply encoded by using the positions of their centroids and enumerating the pixels on their respective bounding contours. However, the raw contour pixels cannot be used directly as the tracked/predicted variables in the Kalman filter model for two reasons:

- 1. The number of contour pixels for a single knot defect may vary significantly in different image slices. This makes formulation of the model a difficult task.
- 2. It is computationally inefficient to use too many tracked/predicted variables in the Kalman filter.

In addition, the complexity of geometrical computation for a nonconvex contour is much higher than that for a convex contour. This is an important consideration in a lumber production planning system, where a significant amount of geometrical computation is entailed in determining the contour(s) of intersection of a knot (or the exterior log surface) and the sawing plane. Thus, it is desirable to encode a knot defect or the exterior log boundary in a single CT slice using a convex hull defined by a small number of points. A Bspline contour approximation algorithm can be used to determine the control points of the convex hull. The control points are then used as the tracking parameters in the Kalman filtering model. In this section, the Kalman filter model for 3-D reconstruction of knots and the outer surface of the log is described.

The Kalman filter model for the reconstruction of knots is depicted in Fig. 8 and is described as follows:

- 1. Initially, a knot defect area is detected in a single CT image using the technique described in Sect. 5.1.
- 2. The contour of the knot defect is extracted and its convex hull is computed. *M* B-spline control points are then used to approximate the convex hull.
- 3. The Kalman filter is applied to predict the "velocity" of the knot defect. Here, the term velocity denotes the rate at which the shape of the knot changes across successive CT image slices. Let  $d_x$  denote the velocity component along the x axis and  $d_y$  the velocity component along the y axis. Let s denote the scale parameter in the time interval [t, t + 1] such that s = 0 denotes the fact that the object size is unchanged in the time interval [t, t + 1], s > 0 denotes that the object size





has increased and s < 0 denotes that the object size has shrunk in the same time interval. If at time t, the centroid of the convex hull is given by  $(c_x, c_y)$ , and the velocity by  $(d_x, d_y, s)$ , then for a point (x, y) on the convex hull at time t, its position (x', y') at time t + 1 can be computed using (4).

- 4. The predicted velocity  $(d_x, d_y, s)$  is used to estimate the new position of the predicted convex contour using (4). The updated convex contour is obtained by using the predicted convex contour to initialize the Snakes contour fitting algorithm. The Snakes contour fitting algorithm is used to search for the actual boundary of the knot in the new CT image slice.
- 5. Steps 2, 3, and 4 are repeated until all the CT image slices are processed or the size of the knot is too small to be classified as a valid knot.

Since the above algorithm uses a combination of Kalman filter-based prediction and contour fitting using Snakes, it is termed as the Kalman Snakes algorithm. The Kalman Snakes algorithm for reconstruction of the exterior log surface is similar, except for some differences in how the convex hull points are generated. The algorithm for convex hull generation is given in the following subsection. The details of the B-spline contour approximation algorithm and Kalman filter-based prediction can be found in [55].

# 6.1.1 Convex hull extraction

Given N 2-D contour points, there exists a subset consisting of M of the N contour points such that these M contour points define a convex contour and each of the original N contour points is either enclosed by this convex contour or is a vertex of the convex contour. The M contour points described above constitute the convex hull of the original N contour points. Graham's algorithm [53] allows for the efficient computation of the convex hull. Graham's algorithm can be briefly described as follows.

The top most point from the original point set is chosen as the starting point. All other points in the original point set are sorted in clockwise order based on their angles formed with the topmost point.Graham's algorithm iteratively adds a new point  $p_i$  from the sorted point array in to the convex point set; and checks whether the point  $p_{i-1}$  added prior to  $p_i$  violates the convexity relationship with respect to points  $p_i$  and  $p_{i-2}$  where  $p_{i-2}$  is the point added prior to point  $p_{i-1}$ . Point  $p_{i-1}$  is deleted if it violates the convexity relationship. The details of Graham's algorithm can be found in [53]. Figure 9 depicts an example of a set of the original points where point  $p_0$  is chosen as the starting point. The other points are sorted and sequentially added to the convex point set. Points  $p_2$ and  $p_7$  are removed because they do not comply with the convexity relationship. The complexity of Graham's algorithm is  $O(N \log N)$ . In order to reduce the computational complexity, we resample

the contour points (increase the sampling interval) in order to reduce N before Graham's algorithm is applied.

It is empirically observed that the error resulting from approximating the contour of a knot defect in a single CT image slice with its convex hull is relatively insignificant since most knot contours are very close to being convex in shape. The outer boundary of a log cross-section, as it appears in a single CT image slice, is also close to being a convex contour in shape. However, since the outer boundary is large, there is a possibility of treating large void areas as normal wood when approximating a slightly nonconvex outer boundary by its convex hull. Therefore, it is necessary to check the difference between the original outer boundary and its convex hull approximation using the following procedure. Let P denote an ordered sequence of the contour points belonging to the outer boundary of the log cross-section. The contour points in P that belong to the convex hull are marked. All possible subsequences in P, such that the endpoints of each subsequence are convex hull points and all other points in the subsequence are nonconvex hull points, are generated. Each such subsequence of points defines a region which is included in the convex hull but does not belong to the interior of the actual contour. The convex hull of each such region, with an area above a predefined threshold, is also stored as part of the contour data in addition to the convex hull of the actual contour. For example, consider the sequence of points  $p_0, p_1, \dots, p_8$ in Fig. 9. If the original order of the contour points is  $p_0, p_1, p_2, p_3, p_4, p_5, p_6, p_7, p_8$ , then based on the aforementioned algorithm, the subsequences that need to be stored are  $p_1, p_2, p_3$  and  $p_6, p_7, p_8$ (Fig. 10). On the other hand, if the original order of the points is  $p_0, p_1, p_3, p_2, p_4, p_5, p_6, p_7, p_8$ , then based on the aforementioned algorithm, the subsequences that need to be stored are  $p_4, p_2, p_3$  and  $p_6, p_7, p_8$  (Fig. 11).

# 6.1.2 Active contour modeling using Snakes

Snakes is an active contour modeling method used to represent and extract deformable contours in an image. Given an initial contour, Snakes evolve to the actual contour by minimizing an energy function, such as the one given in (5). Usually, the Snakes energy function is a sum of two terms, the first representing the internal energy and the second the external energy. The internal energy term imposes a smoothness constraint on the object shape whereas the external energy terms pull the points on the Snakes contour towards the actual object boundary. The internal energy term, which includes the sum of the distances between successive Snakes contour points and the sum of the second-order differences of the Snakes contour points, ensures smoothness and compactness of the object shape. The external energy term in (5) contains the image gradient, which tends to pull the Snakes contour points towards the image pixels with high gradient values.

$$E_{s} = \sum \alpha |\dot{x}(s)| + \beta |\ddot{x}(s)| + E_{ext}(x(s))$$
  
= 
$$\sum \alpha |v_{i} - v_{i-1}| + \beta |v_{i-1} - 2v_{i} + v_{i+1}| + \gamma |\nabla I(v_{i})|$$
  
internal energy (5)

If the image gradient values are very low in the vicinity of the Snakes contour points, there are no external forces to pull the Snakes contour points towards the edge points. Hence, in this situation, the Snakes contour points, although smooth, provide a very poor fit to the actual edge pixel data. For example, the interior of knot defects is usually homogeneous and strong image gradient values exist only along the outer boundary of the knots. Thus, a Snakes contour that is initialized with image pixels within the interior of a knot would not converge on the outer boundary of the knot. In a more complicated situation, a small image gradient value within the interior of the knot could



Fig. 9 An example to illustrate Graham's convex hull finding algorithm



**Fig. 10** Order of the original contour points *p*<sub>0</sub>, *p*<sub>1</sub>, *p*<sub>2</sub>, *p*<sub>3</sub>, *p*<sub>4</sub>, *p*<sub>5</sub>, *p*<sub>6</sub>, *p*<sub>7</sub>, *p*<sub>8</sub>



**Fig. 11** Order of the original contour points *p*<sub>0</sub>, *p*<sub>1</sub>, *p*<sub>3</sub>, *p*<sub>2</sub>, *p*<sub>4</sub>, *p*<sub>5</sub>, *p*<sub>6</sub>, *p*<sub>7</sub>, *p*<sub>8</sub>

draw the Snakes contour points towards the interior of the knot from which it may never recover. There are many methods proposed to tackle the Snakes contour fitting problem within homogeneous image regions. Xu and Prince [54] have proposed a gradient vector flow technique, which iteratively propagates the gradient value from a pixel with high gradient value to the neighboring pixel locations, thus producing a gradient vector flow. This technique solves the problem caused by the presence of locally homogeneous regions when performing Snakes contour fitting, but is computationally intensive as several iterations are needed to establish the vector flow over the entire image. Moreover, when a hole exists within the interior of a knot resulting in strong gradient values around the hole, the gradient vector flow algorithm only makes the situation worse.

A knot region in a single CT image slice can be extracted by using a binarization technique, which allows for a much simpler solution to the problem caused by the presence of homogeneous regions in the image. If a point belongs to the interior of the knot, a force can be designed to move this point away from the centroid of the knot thus causing it to approach the actual outer boundary of the knot. If a point does not belong to the knot, a force can be designed to move this point towards the centroid of the knot thus bringing it closer to the outer boundary of the knot. The redesigned external energy equation  $E_{\text{ext}} =$  $sign(T_k - D(v_i))|v_i - C|$  has this property. Here C is the centroid of the knot,  $D(v_i) = 1/M^2 \sum_{(x,y) \in W(v_i)} F(x,y)$ is the grayscale density at Snakes contour point  $v_i$ measured in a local window  $W(v_i)$  of size  $M \times M$ centered at  $v_i$ , F(x,y) is the image intensity function,  $T_k$ is the binarization threshold for knots and sign(x) = 1 if  $x \ge 0$ ; and sign(x) = -1 otherwise. If  $D(v_i) \ge T_k$ , then  $v_i$  is a knot point and larger  $|v_i - C|$  values will result in lower energy since  $sign(T_k - D(v_i)) = -1$ . On the other hand, if  $D(v_i) < T$ , then  $v_i$  is not a knot point and smaller  $|v_i - C|$  will result in lower energy since  $sign(T_k - D(v_i)) = 1$ . Note that the computation of the

grayscale density  $D(v_i)$  is fairly robust to the presence of small holes within the interior of the knot as long as the overall area covered by the holes within the window  $W(v_i)$  is a small fraction of the window size  $M^2$ . The redesigned energy function can be expressed as follows:

$$E_{s} = \sum \underbrace{\alpha |v_{i} - v_{i-1}| + \beta |v_{i-1} - 2v_{i} + v_{i+1}|}_{\text{internal energy}} + \underbrace{\text{sign}(T_{k} - D(v_{i}))|v_{i} - C|}_{\text{external energy}}$$
(6)

in which  $T_k$  is the chosen binarization threshold for knots,  $D(v_i)$  is the grayscale density value at Snakes contour point  $v_i$ , and C(x,y) is the centroid of the Snakes contour which is a good approximation to the centroid of the knot. This redesigned Snake contour fitting technique is termed as *Binary Snakes* fitting in this paper.

The Binary Snakes fitting technique can be used to extract both knots and the outer log cross-section boundaries in a single CT image slice. The only difference is that in the case of the outer log boundaries the binarization threshold is much smaller than the one used for knots and the window size M can be just 1 since the grayscale difference between the interior and exterior of the log cross-section is typically quite significant. Figure 12a,b depict the results of Binary Snakes contour fitting for knot extraction on two CT image slices.

# 6.1.3 Kalman Snakes

Ì

The Snakes contour fitting algorithm converges to the actual boundary in much fewer iterations if the initial Snakes contour points are chosen close enough to the actual boundary. Therefore it is natural to combine the Kalman filtering algorithm with the Snakes fitting algorithm, where the Kalman filter is used to predict the initial locations of the Snakes contour points in the new CT image slice given the locations of the boundary points in previous CT image slices. A hybrid velocity-and contour-based Kalman filter model proposed by Koller et al. [56, 57] is used in this paper. The model can be described using the following equations:

$$\xi_{k+1}^{-} = \xi_{k}^{+} + q_{k} \tag{7}$$

$$Z_k = H_k \xi_k^- + \nu_k \tag{8}$$

$$X_{k+1}^{-} = X_{k}^{+} + H_{k}\xi_{k} \tag{9}$$

$$Z_k^X = X_k^- + \nu_k \tag{10}$$





Equation (7) represents the velocity transition function, (8) the velocity measurement function, (9) the contour transition function, and (10) the contour measurement function. Each of the two pairs of (7) and (8), and (9) and (10) constitute a standard linear Kalman filter model. Variables with the superscript "–" denote the prior estimations or predicted values. Variables with superscript "+" denote the post estimations or the predicted values after adjustments. The various variables in (7)–(10) are explained as follows:

- $\xi_k$ : The velocity vector at time k.  $\xi_k = (d_x d_y s)^T$ where  $d_x$  is the speed or velocity component along the x axis,  $d_y$  the speed or velocity component along the y axis and s is the scale parameter as described in (4).
- $q_k$ : Random additive noise in the velocity transition function. It is modeled as zero-mean Gaussian white noise that follows the distribution  $\mathcal{N}(0, Q_k)$  A time-invariant (i.e., stationary) noise model is assumed, i.e.,  $Q_k = Q_0$ .
- $v_k$ : Random additive noise in the velocity measurement function. It is modeled as zeromean stationary Gaussian white noise, i.e.,  $v_k \sim N(0, R_k)$  where  $R_k = R_0$ .
- $X_k$ : The points along the contour of an object at time k.  $X_k = (x_1, y_1, ..., x_n, y_n)^T$ , which represents the vector of the B-spline control points in Kalman Snakes model.
- $H_k$ : The measurement matrix given by

$$H_{k} = \begin{pmatrix} I_{2} & (\hat{X}_{1,k}^{+} - C_{k}) \\ \vdots & \ddots \\ I_{2} & (\hat{X}_{n,k}^{+} - C_{k}) \end{pmatrix} \text{ where } I_{2} = \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}.$$

The centroid of the object at time k.

 $Z_k^X$ : The control points of the actual contour obtained from the Snakes fitting algorithm.

 $Z_k$ : The actual velocity measurement used for velocity predictiongiven by  $Z_k = Z_k^X - X_k^-$ .

The Kalman filtering algorithm can be explained as follows:

- 1. Given the adjusted velocity prediction (postprediction)  $\xi_k^+$  at time *k*, predict the velocity at time k + 1 using  $\xi_{k+1}^- = \xi_k^+ + q$  where initially,  $\xi_0^+ = (0 \ 0 \ 0)^T$ .
- 2. Predict the positions of the contour control points  $X_{k+1}^-$  using  $\xi_{k+1}^-$ :  $X_{k+1}^- = X_k^+ + H_k^+ \xi_{k+1}^-$ .
- 3. Compute the prior-covariance of the velocity transition function:  $\pi_{k+1}^- = \pi_k^+ + Q$ .
- Based on the predicted contour control points, compute the actual contour control points Z<sub>k+1</sub> at time k + 1 using the Snakes fitting technique and compute the error as Z̃<sub>k+1</sub> = Z<sub>k+1</sub> X<sup>-</sup><sub>k+1</sub>.
   Compute the postcovariance for the transition
- 5. Compute the postcovariance for the transition function as  $(\pi_{k+1}^+)^{-1} = H_{k+1}' R^- H_{k+1} + (\pi_{k+1}^-)^{-1}$  where *H'* is the transpose of *H*.
- 6. Compute the postcovariance of the measurement function:  $(\Pi_{k+1}^+)^{-1} = R^{-1} + (\Pi_{k+1}^-)^{-1}$ .
- Compute the postprediction for the velocity: *ξ*<sup>+</sup><sub>k+1</sub> = π<sup>+</sup><sub>k+1</sub>(H'<sub>k+1</sub>R<sup>-1</sup> Ž<sub>k+1</sub> + π<sup>-1</sup><sub>k+1</sub> ξ<sup>-</sup><sub>k+1</sub>).
   8. Compute the postprediction for the contour con *ξ*<sup>+</sup><sub>k+1</sub> = π<sup>+</sup><sub>k+1</sub>(H'<sub>k+1</sub>R<sup>-1</sup> Ž<sub>k+1</sub> + π<sup>-1</sup><sub>k+1</sub> ξ<sup>-</sup><sub>k+1</sub>).
   *ξ*<sup>+</sup><sub>k+1</sub> = π<sup>+</sup><sub>k+1</sub>(H'<sub>k+1</sub>R<sup>-1</sup> Z<sub>k+1</sub> + π<sup>-1</sup><sub>k+1</sub> + π<sup>-1</sup><sub>k</sub>
- 8. Compute the postprediction for the contour control points:  $\hat{X}_{k+1}^+ = \Pi_{k+1}^+ (R^{-1}\tilde{Z}_{k+1} + (\Pi_{k+1}^-)^{-1} \hat{X}_{k+1}^-).$

For further details on the Kalman filter, the interested reader is referred to [55].

6.2 Three-dimensional construction of holes and cracks

The Kalman filter-based model for 3-D reconstruction of cracks and holes is given by the following equations:

$$\hat{\xi}_{k+1}^{-} = \hat{\xi}_{k}^{+} + q_{k} \tag{11}$$

$$Z_k = \hat{\xi}_k^- + \nu_k \tag{12}$$

where  $\hat{\xi}$  is the tracking variable and  $Z_k$  is the measured value of the tracking variable. The contour of a hole is encoded by its bounding rectangle. The velocity of the center point of the rectangle  $\xi = (\xi_x, \xi_y)^T$  is used as the tracking variable in the tracking model for holes. Given a prediction  $\tilde{\zeta}_{k+1}^-$ , the new center point of the hole is computed. A rectangular image region centered on the predicted center point is then subject to binarization using (3) and the new location of the hole is obtained exactly using the technique for hole detection in a single CT image slice. Thus, the 3-D reconstruction of a hole is achieved by tracking the hole region across successive CT image slices. The need for explicit correspondence analysis between hole regions extracted independently in successive CT image slices is thus averted.

The orientation angle  $\theta$  of the line segment representing a crack is used as the tracking variable  $\xi$  in the Kalman filter-based tracking model for a crack. After a crack is detected and extracted in the previous CT image slices, its orientation angle in succeeding CT image slices is predicted using (11). A fast crack localization scheme is used to extract the crack in succeeding CT image slices using the predicted orientation. For continuous crack detection, the image is first binarized using the scheme described in [52]. The crack detected in the previous CT image slice is represented by the parameters ( $\rho$ ,  $\theta$ ,  $x_0$ ,  $y_0$ , l) where ( $x_0$ ,  $y_0$ ) is the crack center and l the length of the line segment describing the crack. The crack line is described by the equation  $x \cos \theta + y \sin \theta = \rho$ . The actual orientation angle of the crack in a succeeding CT image slice may be slightly different from the predicted value of  $\theta$ . The following search algorithm is used to determine the value of parameter  $\theta$ .

For each predicted value of  $\theta$ , a rectangle with height l, width  $\delta$  and orientation  $\theta$  and centered at point (x, y) is generated as shown in Fig. 13a. Note that  $\delta$  is a predetermined small value. The image in the interior of the rectangle is projected along orientation  $\theta$ , as shown in Fig. 13b, c. The projected values (the sum of the pixel graylevel values) along the orientation  $\theta$  are stored in array p(x) where  $0 \le x < \delta$ . Let  $f(\theta) = \max_{0 \le x < \delta} p(x)$ . A local search is performed in the range  $[\theta - \Delta \theta, \theta + \Delta \theta]$  and the optimal value  $\theta^*$  is chosen such that  $\theta * = \arg \max_{\theta' \in [\theta - \Delta\theta, \theta + \Delta\theta]} f(\theta')$ . Given the new orientation  $\theta^*$  and  $x^* = \arg \max_{0 \le x \le \delta} p(x)$ , the line corresponding to the updated crack can be deter- $(\rho', \theta') = (x_0 \cos \theta * + y_0 \sin \theta * - x', \theta *)$ mined as (Fig. 14). Next, the center of the crack line and its length are updated using the following procedure. A new rectangle with parameters  $(l_1, \delta_1, x_1, y_1, \theta^*)$  is chosen as shown in Fig. 15a where  $l_1$  is the length of the new rectangle such that  $l_1 > l$ ,  $\delta_1$  is its width, and  $(x_1, y_1)$  is its center point which is the projection of point  $(x_0, y_0)$  on the line  $(\rho', \theta')$  such that  $(x_1, y_1) =$  $(\rho' \cos \theta' - \rho_0 \sin \theta', \rho' \sin \theta' + \rho_0 \cos \theta')$ , where  $\rho_0$ =  $x_0 \sin \theta' - y_0 \cos \theta'$ . Next, the image within the interior of this rectangle is projected along the direction perpendicular to line  $(\rho', \theta')$  as shown in Fig. 15b. The largest continuous area in the projection corresponds to the new crack. However, in some cases, the crack could be broken, therefore all the segments with length larger than a threshold, where the threshold is

Fig. 13 Crack localization in the new slice: **a** rectangle defining the crack, **b** image of the crack in the interior of the rectangle, and **c** projected pixel graylevel values along the orientation  $\theta$ 





Fig. 14 Updating the crack parameters in the new slice



Fig. 15 Determination of the crack length in the new slice: **a** Rectangle defining the crack in the new slice, and **b** projection of the pixel graylevel values along a direction perpendicular to the crack line

chosen to be larger than the gap between the annular rings, are connected together to localize the crack. The two endpoints  $(x_1, y_1)$  and  $(x_2, y_2)$  of the line segment can be obtained from the locations of the two endpoints of the projection. Thus, the center point and the length of the updated crack can be computed and the updated parameters of the crack in the new CT image slice obtained.

The Kalman filter-based tracking method described above for the detection and 3-D reconstruction of crack defects is much faster than detecting crack defects independently in individual CT image slices and then computing the 3-D reconstruction by establishing defect correspondence across the CT image slices. This is so because in the Kalman filter-based tracking approach, it is not necessary to perform all the steps involved in crack detection and localization (i.e., thinning the binarized image, extracting the fork points, grouping the fork points, performing RAN-SAC-based line fitting to the grouped fork points, and performing a search to determine the crack length) in subsequent CT image slices once they have been performed for the first CT image slice in the image sequence. Moreover, by performing the projection to obtain the length of the cracks, some false cracks can be removed if their length is too short.

After the defect detection and localization is performed in the new CT image slice using the Kalman filter-based tracking method described above, the defect areas that are detected and localized are removed from further consideration. The procedures for defect detection and localization in a single CT image slice (i.e., without defect tracking) are applied to the remainder of the image to search for new defects.

# 6.3 Removal of false defects and insertion of missing defects

Spatial coherence is exploited to remove false defects and account for missing defects in the CT image slices. If a defect is detected only in one CT image slice, and no corresponding defect is detected in k previous or ksucceeding CT image slices, where k is a predetermined threshold (in our case k = 2), then the defect is deemed to have been caused by random noise and is removed from further consideration. Likewise, when the size of a defect is small, it is possible that it is not detected in a single CT image slice. If a defect is detected in CT image slices i - 1 and i + 1 but not in CT image slice *i*, then it is necessary to verify whether or not the defect exists in CT image slice *i*. The approximate position of the defect in CT image slice i is computed via interpolation between its positions in CT image slices i - 1 and i + 1. The existence of the defect at the interpolated position in the CT image slice is verified by using the defect detection and localization procedure for a single CT image slice (i.e., without tracking) but with relaxed threshold values. For example, in the case of a hole, the detection and localization procedure described in Sect. 5.3 is used but the value of threshold  $T_h$  in (3) is chosen to be slightly higher. If a hole is detected with the relaxed threshold value, then it is deemed to be a real hole.

The aforementioned techniques for removal of false defects and insertion of missing defects by exploiting spatial coherence have been verified using the available CT image data sets. It was empirically observed that detection of false knots or insertion of missing knots is typically not an issue since knots tend to be fairly large and distinct. Holes, on the other hand, could be missed if they are small in diameter. However, after exploiting spatial coherence it was possible to restore missing hole defects. In the case of crack defects, the Kalman filter-based tracking technique described previously was observed to be robust enough to detect, localize, and compute a 3-D reconstruction of a crack, once detected and localized in previous CT image slices. In the case of false cracks and holes, verifying spatial support from previous and/or succeeding CT image slices was found to be very effective in their removal.

## 7 Experimental results

The proposed Kalman filter-based tracking scheme for detection, localization, and 3-D reconstruction of internal defects in hardwood logs from CT image data was subject to experimental verification and validation. Experiments were conducted on four sets of log data, from three popular hardwood species found in the United States, namely, white ash, red oak, and hard maple, and were labeled as Ash1, Ash2, Maple, and Oak, respectively. The cross-sectional CT images of the hardwood logs were captured using a Toshiba TCT 20AX CT scanner (a third generation CT scanner) with a pixel resolution of 0.75 mm by 0.75 mm, an intensity resolution of 8 bits per pixel (i.e., 256 gray levels) and an image size of  $316 \times 316$  pixels. The scanning of a 4 m log resulted in 224 cross-sectional CT images. All the programs were run on a 2.0 GHz Pentium 4 Xeon workstation with 1.5 GB RAM and 1.0 MB of cache memory.

Figure 16 shows the results of Kalman filter-based tracking and contour fitting using Binary Snakes for detection and localization of knots over a continuous sequence of CT image slices. Figure 17 depicts the resulting 3-D reconstruction of the knots. Likewise, Fig. 18 shows the results for detection and localization of cracks and holes over a sequence of CT images. Holes are represented by their bounding rectangles. Rectangles with same color in different image slices correspond to the same hole. A crack is modeled as a line, but for the sake of clarity a rectangle is used to mark its locations (Fig. 18). A semitransparent view of the log showing its internal defects is depicted in Fig. 19.



Fig. 17 Three-dimensional reconstruction of the knots

Table 1 summarizes the defect detection performance of the proposed scheme for over 224 crosssectional CT image slices of hardwood log data Ash1 for each of the three major internal defect types: knot, hole, and crack. The detection rate, false positive rate, and false negative rate of the proposed scheme are computed by comparing the results of the proposed scheme with those obtained from a human expert grader examining the physically sawn lumber. Also, the performance of the previous scheme [13, 14] that detected and localized defects in each CT image slice independently (i.e., slice by slice without tracking the defects across multiple CT image slices) was compared with the performance of the proposed scheme (Table 1). Although the Kalman filter-based tracking scheme did not result in any improvement in the detection rate or the false positive rate for knots, it did improve the detection rate for cracks from 94 to 98% and the false positive rate from 12 to 2% when compared to the scheme that processed and analyzed each CT image slice independently. Likewise, the proposed Kalman filter-based scheme did not improve the detection rate for holes but did improve the false positive rate from 12 to 3%. Tables 2, 3, and 4 summarize the performance of the proposed Kalman filter-based



Fig. 16 Results of Kalman filter-based tracking and contour fitting using Binary Snakes contour fitting for detection, localization, and 3-D reconstruction of knots

internal defects



Fig. 18 Results of Kalman filter-based tracking and contour fitting using Binary Snakes contour fitting for detection, localization, and 3-D reconstruction of cracks and holes



tracking scheme for hardwood log data Ash2, Maple, and Oak, respectively. In all cases, it can be seen that the proposed Kalman filter-based tracking approach does improve upon the performance of the scheme that processes and analyzes each CT image slice independently in terms of both, defect detection rate and false positive rate.

Table 5 compares the average processing time per CT image slice of the proposed Kalman filter-based tracking scheme and the scheme that processes and analyzes each CT image slice independently (i.e., slice by slice) on the hardwood log data Ash1. In order to ensure a fair comparison, the detection time for the outer log boundary and each of the three major internal defect types, i.e., knot, crack, and hole are measured and tabulated independently. Note that the 3-D defect reconstruction in the proposed method is done automatically by using the integrated detectiontracking scheme, whereas the slice-by-slice scheme needs extra time to compute the defect correspondences in order to reconstruct the 3-D defect model. The time used to compute the defect correspondences varies with the different number of defects and the methods used to compute the correspondences as tabulated in our previous work [13]. Note that the time taken to compute the defect correspondences is not

**Table 1** The detection rate for the three major defect types on log data Ash1

Detection rate	Knot (total 24)			Crack (total 112)			Hole (total 25)		
	Correct detection	False negatives	False positives	Correct detection	False negatives	False positives	Correct detection	False negatives	False positives
Slice-by-slice Tracking	24 (100%) 24 (100%)	$\begin{array}{c} 0 \ (0\%) \\ 0 \ (0\%) \end{array}$	0 (0%) 0 (0%)	105 (94%) 110 (98%)	7 (6%) 4 (4%)	13 (12%) 2 (2%)	24 (96%) 24 (96%)	1 (4%) 1 (4%)	3 (12%) 1 (4%)

Table 2 The detection rate for the three major defect types on log data Ash2

Detection rate	Knot (total 135)			Crack (total 22)			Hole (total 159)		
	Correct detection	False negatives	False positives	Correct detection	False negatives	False positives	Correct detection	False negatives	False positives
Slice-by-slice Tracking	135 (100%) 135 (100%)	$\begin{array}{c} 0 \ (0\%) \\ 0 \ (0\%) \end{array}$	0 (0%) 0 (0%)	21 (95%) 21 (95%)	1 (5%) 1 (5%)	3 (15%) 1 (5%)	159 (100%) 159 (100%)	0 (0%) 0 (0%)	0 (0%) 0 (0%)

Detection rate	Knot (total 12)			Crack (total 21)			Hole (total 27)		
	Correct detection	False negatives	False positives	Correct detection	False negatives	False positives	Correct detection	False negatives	False positives
Slice-by-slice Tracking	12 (100%) 12 (100%)	0 (0%) 0 (0%)	0 (0%) 0 (0%)	19 (90%) 21 (100%)	2 (10%) 0 (0%)	1 (5%) 1 (5%)	27 (100%) 27 (100%)	0 (0%) 0 (0%)	0 (0%) 0 (0%)

Table 3 The detection rate for the three major defect types on log data Maple

Table 4 The detection rate for three major defect types on log data Oak

Detection rate	Knot (total 15)			Crack (total 0)			Hole (total 15)		
	Correct detection	False negatives	False positives	Correct detection	False negatives	False positives	Correct detection	False negatives	False positives
Slice-by-slice Tracking	15 (100%) 15 (100%)	0 (0%) 0 (0%)	0 (0%) 0 (0%)	0 (0%) 0 (0%)	0 (0%) 0 (0%)	0 (0%) 0 (0%)	15 (100%) 15 (100%)	0 (0%) 0 (0%)	0 (0%) 0 (0%)

 $\label{eq:table_$ 

	Boundary	Knots	Crack	Hole
Slice-by-slice	40	95	101	15
Tracking	16	61	81	15

included for the slice-by-slice method in Table 5, whereas in the case of the integrated detection-tracking scheme, the 3-D model is established automatically.

# 8 Conclusions and future work

In this paper, a novel feature-based tracking approach was proposed and implemented for the detection, localization, and 3-D reconstruction of internal defects in hardwood logs from cross-sectional CT images. In contrast to traditional methods, where the defects are detected and localized independently in individual CT image slices and the 3-D reconstruction of the defects accomplished via correspondence analysis across the various CT image slices, the proposed system integrated defect detection, defect localization, and 3-D defect reconstruction by incorporating a Kalman filterbased feature tracking scheme. The defects were simultaneously detected, classified, localized, and reconstructed in 3-D space making the proposed scheme computationally much more efficient than existing methods. Robust techniques for defect detection and classification were proposed and implemented. Defect class-specific tracking schemes based on the Kalman filter were designed which use the geometric parameters of the defect contours as the tracking variables. The geometric parameters of the defect contours are computed using a combination of B-spline contour approximation and an improvised Binary Snakes contour fitting procedure termed as Kalman Snakes. Robust techniques for extraction and characterization of the external log surface were also designed. Experimental results on cross-sectional CT images of hardwood logs from select species such as white ash, hard maple, and red oak demonstrated that the proposed scheme was a significant improvement over the traditinal schemes that processed and analyzed each CT image slice independently in terms of both detection accuracy and processing speed. The results of defect detection, defect localization, and 3-D defect reconstruction could be used for automatic planning of log sawing and lumber production strategies, and as an interactive graphical tool for the training of novice sawyers. These research directions will be pursued in the near future.

Acknowledgments This research was supported in part by a NRICGP award by the US Department of Agriculture to Drs. Bhandarkar, Daniels, and Tollner (Award Number 2001-35103-10049).

# References

- 1. Haygreen JG, Bowyer JL (1996) Forest products and wood science: an introduction, 3rd edn. Iowa State University Press, Ames, Iowa
- Hodges DG, Anderson WC, McMillin CW (1990) The economic potential of CT scanners for hardwood sawmills. For Prod J 40(3):65–69
- 3. Chang SJ (1990) NMR application for internal defect detection. In: Proceedings of process control/production management of wood products: technology for the 90s, Athens, Georgia

- Conners RW (1987) A computer vision system for grading hardwood lumber. In: Proceedings of the 2nd international conference on scanning technology in sawmilling, Oakland, CA
- Conners RW (1990) A multisensor machine vision system for hardwood defect detection. In: Proceedings of process control/ production management of wood products: technology for the '90s, Annual Meeting, SE Section FPRS, Athens, GA
- Funck JW (1990) Softwood veneer defect detection using machine vision. In: Proceedings of process control/production management of wood products: technology for the '90s, Annual Meeting, SE Section FPRS, Athens, GA,
- Funt BV, Bryant EC (1987) Detection of internal log defects by automatic interpretation of computer tomography images. For Prod J 37(1):56–62
- Hailey JRT, Swanson JS (1987) Imaging wood using magnetic resonance. In: Proceedings of the 2nd international conference scanning technology in sawmilling, Oakland, CA
- Taylor FW, Wagner FG Jr, McMillin CW, Morgan IL, Hopkins FF (1984) Locating knots by industrial tomography—a feasibility study. For Prod J 34(5):42–46
- Wagner FG, Taylor FW, Ladd DS, McMillin CW, Roder FL (1989) Ultrafast CT scanning of an oak log for internal defects. For Prod J 39(11/12):62–64
- 11. Richards DB (1977) Value yield from simulated hardwood log sawing. For Prod J 27(12):47–50
- 12. USDA Forest Service (1988) The South's fourth forest: alternatives for the future. Forest service report no. 24
- Bhandarkar SM, Faust T, Tang M (1999) CATALOG: a system for detection and rendering of internal log defects using computer tomography. J Mach Vis Appl 11:171–190
- Bhandarkar SM, Faust TD, Tang M (2002) Design and prototype development of a computer vision-based lumber production planning system. J Image Vis Comput 20(3):167– 189
- 15. Roder FL, Scheinman E, Magnuson P (1989) High-speed CT scanning of logs. In: Proceedings of the 3rd international conference on scanning technology in sawmilling, San Francisco, California
- 16. Kalman RE (1960) A new approach to linear filtering and prediction problems. ASME J Basic Eng, Ser D: 35–45
- Chang SJ, Olson JR, Wang PC (1989) NMR imaging of internal features in wood. For Prod J 39(6):43–49
- McMillin CW (1982) Application of automatic image analysis to wood science. Wood Sci 14(3):97–105
- Forrer JB, Butler DA, Funck JW, Brunner CC (1988) Image sweep-and-mark algorithms: part 1, basic algorithms. For Prod J 38(11/12):75–79
- Forrer JB, Butler DA, Brunner CC, Funck JW (1988) Image sweep-and-mark algorithms: part 2, performance evaluations. For Prod J 39(1):39–42
- McMillin CW, Conners RW, Huber HA (1984) ALPS—a potential new automated lumber processing system. For Prod J 34(1):13–20
- 22. Butler DA, Brunner CC, Funck JW (1989) A dual-threshold image sweep-and-mark algorithm for defect detection in veneer. For Prod J 39(5):25–28
- Li P, Abbot AL, Schmoldt DL (1996) Automated analysis of CT images for inspection of hardwood logs. Proc. IEEE Int. Conf. Neural Netw: 1744–1749
- Schmoldt DL, Li P, Abbot AL (1997) Machine vision using artificial neural networks with local 3D neighborhoods. Comput Electron Agric 16:255–271

- 25. Schmoldt DL, He J, Abbot AL (1998) A comparison of several artificial neural network classifiers for CT images of hardwood logs. In: Proceedings of the SPIE conference machine vision applications in industrial inspection VI, vol 3306, pp 34–43
- Butler DA, Brunner CC, Funck JW (1993) An adaptive image preprocessing algorithm for defect detection in Douglas-fir veneer. For Prod J 43(5):57–60
- Coates ER, Chang SJ, Liao TW (1998) A quick defect detection algorithm for magnetic resonance images (MRI) of hardwood logs. For Prod J 48(10):68–74
- 28. Sarigul E, Abbot AL, Schmoldt DL (2003) Progress in analysis of computed tomography (CT) images of hardwood logs for defect detection. In: Proceedings of the tenth international conference on scanning technology and process optimization in the wood industry, ScanTech, Seattle, Washington, November 2003, pp 19–30
- Zhu D, Conners RW, Schmoldt DL, Araman PA (1996) A prototype vision system for analyzing CT imagery of hardwood logs. IEEE Trans. Syst Man Cybern Part B: Cybern 26(4):522–532
- Samson M (1993) Modeling of knots in logs. Wood Sci Technol 27:429–437
- Davis J, Wells P, Benci N, Grant J, McLachlan D (1995) Portable X-ray CT scanner for electricity power pole and tree inspection and a high resolution laboratory CT system. In: Proceedings of the 2nd international workshop on scanning technology and image processing on wood, Skelleftea, Sweden, pp 9–22
- 32. Grundberg S, Grönlund A (1995) The development of a log scanner for Scots Pine. In: Proceedings of the 2nd international workshop on scanning technology and image processing on wood, Skelleftea, Sweden, pp 40–50
- 33. Som S, Svalbe I, Davis J, Grant J, Gold E, Tsui K, Wells P (1995) Internal scanning of logs for grade evaluation and defect location. In: Proceedings of conference on digital image computing: techniques and applications (DICTA), Brisbane, Australia, pp 408–413
- 34. Svalbe I, Som S, Davis J, Grant J, Tsui K, Wells P (1995) Internal scanning of logs for grade evaluation and defect location. In: Proceedings of the 2nd international workshop on scanning technology and image processing on wood, Skelleftea, Sweden, pp 1–8
- 35. Chiorescu S, Grönlund A (2000) Validation of a CT-based simulator against a sawmill yield. For Prod J 50(6):69–76
- Guddanti S, Chang SJ (1998) Replicating sawmill sawing with TOPSAW using CT images for a full-length hardwood log. For Prod J 48(1):72–75
- Occena LG, Schmoldt DL (1996) GRASP: a prototype interactive graphic sawing program. For Prod J 46(11/12):40– 42
- Schmoldt DL, Li P, Araman PA (1996) Interactive simulation of hardwood log veneer slicing using CT images. For Prod J 46(4):41–47
- 39. Gatchell C, Klinkhachorn P, Kothari R (1992) ReGS—a realistic grading system. For Prod J 32(10):37–40
- Klinkhachorn P, Schwehm CJ, McMillin CW, Huber HA (1989) HaLT: a hardwood lumber training program for graders. For Prod J 39(2):38–40
- Klinkhachorn P, McMillin CW, Kothari R, Yost D (1992) HaLT2: an enhanced lumber grading trainer. For Prod J 42(10):38–40
- 42. Klinkhachorn P, Kothari R, Prasad R, McMillin CW, Mukherjee K (1993) Prototyping an automated lumber processing system (ALPS). For Prod J 43(2):11–18

- Klinkhachorn P, Kothari R, Annavajjhala R, McMillin CW (1993) TRSys—hardwood lumber grading training and remanufacturing system. For Prod J 44(9):69–72
- 44. Moody J, Gatchell CJ, Walker ES, Klinkhachorn P (1998) An introduction to UGRS: the ultimate grading and remanufacturing system. For Prod J 48(9):45–50
- 45. Chang SJ, Guddanti S, Cooper CS (1997) Measuring the benefits of internal log defect scanning: a mill-based study. In: Proceedings of the international conference on scanning technology and process optimization for the wood products industry, November 12–14, 1997, Charlotte, NC, pp 25–28
- Pham DT, Alcock RJ (1998) Automated grading and defect detection: a review. For Prod J 48(4):34–42
- 47. Mairinger F (2003) Strahlenuntersuchung an Kunstwerken. E.A. Seemann, Berlin
- Kass M, Witkin A, Terzopoulos D (1988) Snakes: active contour models. Int J Comput Vis 1(4):321–331
- 49. Sonka M, Hlavac V, Boyle R (1998) Image processing, analysis and machine vision, 2nd edn. PWS Publishing, Boston
- Cychosz JM (1994) Efficient binary image thinning using neighborhood maps. Graphics gems IV. Academic, New York, pp 465–473
- 51. Fischler M, Bolles R (1981) Random sample consensus: a paradigm for model fitting with applications to image analysis and automated cartography. Commun ACM 24(6):381–395
- Bhandarkar SM, Luo X, Daniels R, Tollner EW (2005) Detection of cracks in computer tomography images of logs. Pattern Recognit Lett 26(5):2282–2294
- Graham RL (1972) An efficient algorithm for determining the convex hull of a finite point set. Inf Proc Lett 1(4):132–133
- 54. Xu C, Prince JL (1997) Gradient vector flow: a new external force for Snakes. In: Proceedings of the IEEE conference on computer vision pattern recognition, San Juan, Puerto Rico, pp 66–71
- 55. Koller D, Weber J, Malik J (1993) Robust multiple car tracking with occlusion reasoning. Computer Science Division (EECS), University of California, Report No. UCB/ CSD 93-780, Berkeley, California, USA, October 1993
- 56. Koller D, Weber JW, Malik J (1994) Robust multiple car tracking with occlusion reasoning. In: Proceedings of the European conference on computer vision. Stockholm, Sweden, pp 189–196
- 57. Koller D, Weber J, Huang T, Malik J, Ogasawara G, Rao B, Russell S (1994) Towards robust automatic traffic scene analysis in real-time. In: Proceedings of the international conference on pattern recognition, Jerulasem, Israel, October 1994, pp 126–131

#### **Author Biographies**



**Dr. Suchendra M. Bhandarkar** received a B.Tech. in Electrical Engineering from the Indian Institute of Technology, Bombay, India in 1983, and an M.S. and Ph.D in Computer Engineering from Syracuse University, New York in 1985 and 1989, respectively. He is currently a Professor in the Department of Computer Science at the University of Georgia, Athens, Georgia where he directs the Visual and Parallel Computing Laboratory (VPCL). Dr. Bhandarkar was a Syracuse University Fellow for the academic years 1986-1987 and 1987-1988. He is a member of the IEEE, AAAI, ACM, and SPIE. He is a coauthor of the book Object Recognition from Range Images (Springer Verlag, 1992). He is also a member of the honor societies Phi Kappa Phi and Phi Beta Delta. His research interests include computer vision, pattern recognition, multimedia systems, artificial intelligence and parallel algorithms and architectures for computer vision and multimedia. He has over 100 published research articles in these areas including over 50 articles in peer-reviewed archival journals. Dr. Bhandarkar serves on the editorial boards of the Computer Journal and the Applied Intelligence Journal and on the program committees of several international conferences and symposia. He serves on the technical committees for Pattern Analysis and Machine Intelligence and Multimedia Computing in the IEEE Computer Society and the technical committee for Robot Vision in the IEEE Robotics and Automation Society. He also serves as a consultant to various organizations in the private and public sector including government agencies such as NASA, Department of Defence, Department of Energy, and Department of Health and Human Services.





**Mr. Xingzhi Luo** received a Bachelor's degree from the Department of Automation Control at Huazhong University of Science and Technology, Wuhan, China in 1997, and a Master's degree in the Institute of Automation, Chinese Academy of Sciences. He is currently a Ph.D student in the Department of Computer Science at the University of Georgia, Georgia, USA. He is a student member of IEEE.

Richard F. Daniels graduated with a B.S. in forestry from Rutgers in 1973 and went on to complete a M.S. in forest biometrics at Virginia Tech in 1975. Daniels' PTAEDA model from this time, with Harold Burkhart, was the first individual tree distancedependent growth model for southern pines; versions of this model are still in use today. From 1975 to 1980 he worked as Research Associate on the VA Tech faculty modeling host dynamics and insect suscepti-

bility under the Expanded Southern Pine Beetle Research and Applications Program. Daniels completed his Ph.D. in 1981 and began what would be a 17+ year career with Westvaco Corporation in Summerville, SC, involving several research and forest operations assignments. At Westvaco he was research biometrician, Biometrics Project Leader, Southern Woodlands Operations Manager, and Forest Productivity Research Manager. In 1998 he accepted a post as Professor of Quantitative Forest Management at the University of Georgia's Warnell School of Forest Resources. At Georgia he teaches in the area of silviculture, forest mensuration, modeling, and management. He has established a research program in the area of forest modeling and wood quality. In 2000 he established the Wood Quality Consortium at UGA, a research partnership of eight major forest industry firms and six research institutions with goals to improve predictions of wood quality and management decision tools under various silvicultural practices. Daniels has been active in SAF since 1972, when he was chairman of the Rutgers student chapter in the Allegheny section. In Virginia and SC he was active in the Appalachian section as treasurer and chairman of the Edisto chapter, member of the SC and Appalachian Technical committees, among other service. At Georgia, Daniels serves as faculty advisor to the UGA Student Chapter and has become involved in the local Oconee chapter, Georgia, and Southeast SAF.

**Dr. Ernest W. Tollner** is a native of Maysville, KY and received his BS and MS degrees in agricultural engineering at the University of Kentucky. He did his doctorate at Auburn University. His graduate work was concerned with erosion control, water



resource development and animal waste management. This work provided the foundation for extension into composting and bioconversion research. Research over the past 25 years at the University of Georgia has resulted in two patents, one for a large core soil sampler and the other for a Cone Penetrometer for measuring soil mechanical impedance. A patent for a software package for virtual reconstructive craniofa-

cial surgery from Computer Tomography images is currently pending. His research interests include watershed assessment and modeling, modeling tools for postharvesting, water supply development and pond water balance modeling, sediment control modeling and techniques and X-ray computed tomography for soil property and food quality determination and agricultural and forestry applications.